Author’s response to reviews

Title: Assessment of facility and health worker readiness to provide quality antenatal, intrapartum and postpartum care in rural Southern Nepal

Authors:

Tsering Lama (tplama07@gmail.com)
Melinda Munos (mmunos@jhu.edu)
Joanne Katz (jkatz1@jhu.edu)
Subarna Khatry (skhatry@wlink.com.np)
Steven LeClerq (sleclerq@mos.com.np)
Luke Mullany (lmullany@jhu.edu)

Version: 2 Date: 20 Dec 2019

Author’s response to reviews:

Technical Comments:
1. Please update the Availability of data and materials section to include the DOI for the data.

Response - we have included the doi in the Declarations section; The JHU Data Archive will publish the doi when the paper is published.

2. Please ensure that all figures, tables and supplementary/additional files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Response: all the figures and tables are referenced in the text

3. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.

Response: there are no track changes in the document, and prior cover letters have been removed.

Thank you for your continued consideration of this manuscript.